To
The Editor,
JMCR team,

Dear Sir or Madam,

Re: A case of VACTERL spectrum presenting with Portal Hypertension: a case report

Thank you very much for looking into the revised manuscript. I have gone thorough and made the needful correction as pointed out by the reviewer.

I have listed the issues raised by the reviewer as below:

1. Antenatal and family history: I have updated the details.
2. Spellings and English: I have tried to my best abilities to correct them.
3. Whether the thread portal vein is congenital defect or not: The images were reviewed by the hepato-radiologists. They could not explain the aetiology. We have discussed the possible aetiology of extra hepatic portal venous hypertension and highlighted the need for further evidences.

Once again, many thanks for your expected help,

Sincerely,

Dilli Bhurtel